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Date: September 1, 2004 

KAISER HILL 
C O M P A N Y  

MEMORANDUM 

To: Jan Robbins, T130G 

From: David Ward 

Subject: WEEKLY STATUS MEETING DOULRA for September 1,2004 

Attached are the minutes for the referenced weekly meeting for the Operating Record. 
In attendance were: 

M. Brown 
K. Ferrera 
R. Leitner 
S. McNitt 
D. Onyskiw 

J. Parkin 
D. Rack 
R. Vonfeldt 
D. Ward 

Additional items discussed: 
1. The monthly walkdown for September is completed on September 1,2004 without DOE. 

Actions Items: 

1. Provide DOE with the gram loading for the overmass containers. - R. Leitner. 
2. Contact Warren Seyfe? to determine his status for September and who his back is. - D. 

Ward. 

End 
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B371/374 CLOSURE PROJECT 

WEEKLY STATUS MEETING 
[Every Wednesday; 9:30 am'. In T37121 

AGENDA 
September 1,2004 

INTRODUCTION - David Ward (~5938)  

- D&D - Ken Ferrera (~4436)  

CONSTRUCTION / MAINTENANCE - Dan Coyne (~2820) 

ENVIRONMENTAL & ACTION ITEMS - David Ward 

D&D WASTE - Randy Leitner 

1.  OUTSTANDING ACTION ITEMS 

2. OTHER: I 

The September monthly walkdown did not occur today and will be rescheduled 

a 



Set 
6 lAQ31DR02B 

D&DStatus KenFerrera * I * -  

0 GB 59 removed. 

0 

GB307 removed. 
0 

0 GB 32 removed. 

0 Mechnical isolation in process. 

0 

Glovebox internal component removal is complete for GB 58 and 59. 

D&D of Room 354913547 in process. All tanks removed 
Room 3553,3555,3563, and 3559 set scope complete. 
Completed the removal of all 8 YO’S and GB 83. 
Entries into CSV continue; rail removal complete spraying of racks on west end of CSV complete; 1st vehicle is 
disassembled; 2“1 vehicle disassembly is in process. 
Mechanical isolation and stripout in progress. 
Tank 230B and 131B removal complete. 

Canyon entries, electrical isolation and stripout, and spraying fixative are in process. 
Mechanical isolation, removal, and decQn are complete. 

Started criticality drain removal in basement. 
Plenum 125B complete, filters removed, area deconned and fixed. 
Plenum 242 complete, filters removed, area deconned and fixed. 
Electrical isolation started on Plenum 241. 

Electrical isolation in process for Plenum 141. 
Plenum 141 complete filters removed, deconed, and fixative applied 

-tusDanCoyne . * ”  

9 (CSV) 

10 1 AD1 lOBD00 

17 lADl17BD00 

Environmental 

RCRA stable - no insps. 

Environmental Status 

38 
40 1AD140BD00 
41 1 AD141 BDOO 

. ,  

19 Sludge Removal 

- 
0 

0 

0 

0 

0 

0 

0 

0 

0 

0 

. . 
D&D Summarv ( ACC/ADD) . 

0 .  Pumping wastewater to Eurotainer. 
Piping and electrical removal in process 

Washineton C 
Sets 1 and 2: Electrical and mechanical (pipinglsupports) removal 
Set 6: Dry decon in rooms 355513563 pending. 
Set 12 : Mechanical Removal (room 1103 pipinglchillers) 
Set 13: Perfoming dry decon in room 2317 (pump pads). 
Set 14: On hold 
Set 18: Mechanical removal (tank prep., duct, misc. supports & piping). Prepare for roof 
opening. Cont’d transite removal. 
Set 19: Mechanical removal (tank 875 & GB). 
Set 21: Removing wall btwn room 3801 & 3810. 
Set 22: Cont’d abandoned electrical removal. 
Set 23: Dry decon 3321U341. Prepare for mechanical removal. 

0 Set 26: Complete 

September 1 ,2004 Inventory: 204 drums require handling (10 overmass drums) 

Repackage/disposition the remaining 3000 legacy drums (May 3,2002 starting inventory 3,161 

D&D Waste - 

15 SWB require handling (10 overmass SWBs) 

0 All tanks but 4 are RCRA 
Stable. 

DUD Activities 

0 

0 

0 

0 Performing abandoned electrical removal in room 23 16. . 
0 

Set 33: Electrical removal completed. 
Set 38: Dismantling of SAU-203 placed on hold to re-direct crew to FP-242. 
Set 40: Continued plenum FP-125B removal. 
Set 41: Plenum FP-242 removal. 
Set 57: Continued transite removal and dry decon in room 3810. 
Set 58: Continued removal of GB-118 and GB-120. 

Temporary electrical power installation on-going in both bldgs. 
Outside: Cont’d saw cuttting for airlock at 37 1 basement access. Continued excavation 
for underground piping removal in support of 374 wall opening. Prepare for Set 18 roof 
opening. 

tandy Leitner 
Net Container Loss for Week: Drums - 2 . SWRF-3 
h m s .  
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Date: 

To: 

From: 

Subject: 

KAISER HILL 
C O M P A N Y  

' MEMORANDUM 

August 25,2004 

Jan Robbins, T130G , 

David Ward 

WEEKLY STATUS MEETING DOE/LRA for August 25,2004 

Attached are the minutes for the referenced weekly meeting for the Operating Record. 
In attendance were: 

M. Brown 
D. Coyne 
K. Ferrera 
S. Garcia 
R. Leitner 

S. McNitt 
D. Onyskiw 
D. Ward 

Additional items discussed: 
1. The monthly walkdown for September is tentatively scheduled for next Wednesday 

September 1 at 8:OO a.m. 
2. There is a meeting scheduled for the Wednesday following the Labor Day Holiday, 

September 8. 

Actions Items: 

1. E-mail a copy of the week's status report everyone at the meeting and DOE. 

End 

371/374 Project 
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B371/374 CLOSURE PROJECT 

WEEKLY STATUS MEETING 
[Every Wednesday, 9:30 am. 8371 Conference Room] 

. AGENDA 
August 25,2004 

INTRODUCTION - David Ward (~5938) 

- D&D - Ken Ferrera (~4436) 

CONSTRUCTION / MAINTENANCE - Dan Coyne (~2820) 

ENVIRONMENTAL & ACTION ITEMS - David Ward 

D&D WASTE - Randy Leitner 

1.  OUTSTANDING ACTION ITEMS 

2. OTHER: 

The September monthly walkdown is currently schedule for next week Wednesday September 1, 
at 8:OO a.m. At this time, there are no Sets scheduled for completion walkdowns. 



. 

GB307 removed. 

GB 32 removed. 

Canyon entries, electrical isolation and stripout, and spraying fixative are in process. 
Mechanical isolation, removal, and decon are complete. 

Plenum 125B complete, filters removed, area deconned and fixed. 
Plenum 242 complete, filters removed, area deconned and fixed. 
Electrical isolation started on Plenum 241. 
Electrical isolation and stripout in process for Plenum 141. 
Filter and demister removal complete, decon and spraying fixative in process. 

Construction Status Dan Coyne 

Set 
6 1 AQ3 1 DR02B 

Environmental Status 

9 (CSV) 

10 lADl lOBD00 

17 1 AD1 17BD00 

August 25,2004 Inventory: 206 drums require handling (10 overmass drums) 
17 SWB require handling (1 1 overmass SWBs) 

40 1AD140BD00 
41 1 AD 141 BDOO 

Net Container Loss for Week: Drums - 14 
SWBS - 6 

50 

19 Sludge Removal 

D&D S u m m a r y  ( ACCIADD) 
Weekly Meeting Date 8/25/04 

D&D Status Ken Ferrera 

GB 59 removed. 

0 

Glovebox internal component removal is complete for GB 58 and 59. 

D&D of Room 3549/3547 in process. All tanks removed 
Room 3553,3555,3563, and 3559 set scope complete. 
Completed the removal of all 8 YO’S and GB 83. 
Entries into CSV continue; rail removal complete spraying of racks on west end of CSV complete; 1st vehicle is 
disassembled; 20d vehicle disassembly is in process and the CSV is on air movers. 
Mechanical isolation and stripout in progress. 
Tank 230B and 131B removal complete. 

Environmental 
MRT;IE quarterly insps.; 

RCRA stable -no insps. 

0 

Piping removal 
Electrical removal in progress. 

0 Pumping process waste to Eurotainer 

Washington G 

Set 14: Mechanical stripout. 

Set 21: Removing electricakupports. 

Sets 1 and 2: Electrical and mechanical (pipinglsupports) removal 
Set 6: Dry decon rooms 355513563. 
Set 12 : Mechanical Removal (room 1103 pipinglchillers) 
Set 13:Prepare for dry decon in room 2317. Heat exchanger removal on hold til west 
wall opening is completed. 

Set 18: Mechanical removal (tank platforms, duct, misc. supports & piping). Prepare for 
roof opening. 
Set 19: Mechanical removal (misc. supports). Completed upper wall opening. 

Set 22: Cont’d abandoned electrical removal. 
Set 23: Dry decon 3321U341 is in process . ’  
Set.26 Dry decon in room 3602 is on hold 

D&D Waqte - 

All tanks but 4 have been 
submitted to the State as 
RCRA Stable, 2/27/04 

)up Activities 
Set 33: Electrical removal pending scaffold installation. 
Set 38: Dismantling of SAU-203 placed on hold to re-direct crew to FP-242. 
Set 40: Commenced plenum FP-125B removal. 
Set 41: Plenum FP-242 removal. 
Set 51: Plenum FP-142 removal completed. 
Set 57: Continued transite removal and stair #11 removal. 
Set 58: Continued removal of GB-118 andGB-120 floor. Dry decon 4805. 
Performing abandoned electrical removal in room 23 16. 
Temporary electrical power installation on-going in both bldgs. 
Outside: Cont’d airlock for 371 basement access. Commenced excavation for 
underground piping removal in support of 374 excavation. Transite removal at dock#18 
and asbestos roofing removal for Set 18 opening. Construct temp. cover for Set 18 roof 
opening. 

randy Leitner 
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